Author's response to reviews

Title: Hospitalization for diseases attributable to human papillomavirus in the Veneto Region (North-East Italy)

Authors:

Vincenzo Baldo (vincenzo.baldo@unipd.it)
Sivia Cocchio (sliaciocchio@libero.it)
Alessandra Buja (lessandra.buja@unipd.it)
Tatjana Baldovin (tatjana.baldovin@unipd.it)
Patrizia Furlan (patrizia.furlan.73@alice.it)
Chiara Bertoncello (chiara.bertoncello@unipd.it)
Mario Saia (drmariosaia@gmail.com)

Version: 5 Date: 13 September 2013

Author's response to reviews: see over
Dear Editor,
We have revised the manuscript and would like to re-submit it for your consideration. We look forward to hearing from you soon.
Yours sincerely,
Vincenzo Baldo

Replies to Reviewer 1
none

Replies to Reviewer 2

Discretionary revisions
… I would suggest, if possible to comment more in depth this part and to re-write the second sentence in the last page of discussion without a specific mention to quadrivalent vaccine.

- A comment has been added (page 9, line 14) and the sentence has been changed (page 10; line 6)